Author's response to reviews

Title: Five-year teaching evaluation of oral microbiology experiment curriculum

Authors:

    Min Nie (niemin@whu.edu.cn)
    yan Zhen Gao (973066427@qq.com)
    yu Xin Wu (645933142@qq.com)
    xi Chen Jiang (354454133@qq.com)
    hui Jia Du (137711727@qq.com)

Version: 3  Date: 4 November 2014

Author's response to reviews: see over
Dear Editor,

Thanks for your suggestion about our MS: 1689765945146228 “Five-year teaching evaluation of oral microbiology experiment curriculum”.

We added a paragraph in the Methods:
This entire study was submitted and reviewed by the Medical Ethics Committee of Wuhan University, who subsequently waived the need for ethics. All participants (271 students) were informed of the aims of the quantitative and qualitative evaluation, and consent to participate in the study.

Sincerely,
Nie min
School and Hospital of Stomatology, Wuhan University,
Luoyu Road 237, Wuhan 430079, Hubei, China.
Tel.: +86 27 87686102;
E-mail address: niemin@whu.edu.cn